Author's response to reviews

Title: Primary Central Nervous System Angiosarcoma: Diagnosis and Treatment Strategies

Authors:

James R Hackney (jhackney@uab.edu)
Cheryl A Palmer (capalmer@uab.edu)
Kristen O Riley (koriley@uabmc.edu)
Joel K Cure (jcure@uabmc.edu)
Hassan M Fathallah-Shaykh (hfathall@uab.edu)
Louis B Nabors (bnabors@uab.edu)

Version: 2 Date: 15 November 2011

Author's response to reviews: see over
November 15, 2011

Professor Michael Kidd AM  
c/o Journal of Medical Case Reports  
Flinders University  
GPO Box 2100  
Adelaide 5001  
South Australia

Dear Professor Kidd:

On behalf of my coauthors, I am submitting our case series, “Primary Central Nervous System Angiosarcoma: Diagnosis and Treatment Strategies,” for consideration for publication in *Journal of Medical Case Reports*.

Central nervous system angiosarcomas are rare tumors; our group has been fortunate enough to care for two patients with such tumors recently, and took the opportunity to review the neuro-oncology literature for treatment strategies. Finding little published, we felt a timely review in an open access journal may be helpful to other clinicians who find themselves responsible for the care of these patients.

The manuscript has not been submitted elsewhere for publication. This manuscript has been accepted for presentation at the 16th Annual Society for NeuroOncology Meeting in Orange County, California, November, 2011.

Thank you very much for your time and consideration.

Sincerely yours,

Cheryl Ann Palmer, M.D.  
Division of Neuropathology  
University of Alabama at Birmingham  
1960 6th Avenue South; PD6A 175E  
Birmingham, AL 35294  
(205) 934-2164 – phone  
(205) 975-7548 - fax  
Email: capalmer@uab.edu